Author’s response to reviews

Title: Treatment Selection of Early Stage Non-Small Cell Lung Cancer: The Role of the Patient in Clinical Decision Making

Authors:

Sahar Mokhles (s.mokhles@erasmusmc.nl)
Joost Nuyttens (j.nuyttens@erasmusmc.nl)
Mark de Mol (MdeMol1@amphia.nl)
Joachim Aerts (j.aerts@erasmusmc.nl)
Alex Maat (a.p.w.m.maat@erasmusmc.nl)
Ozcan Birim (o.birim@erasmusmc.nl)
Ad Bogers (a.j.j.c.bogers@erasmusmc.nl)
Hanneke Takkenberg (j.j.m.takkenberg@erasmusmc.nl)

Version: 4 Date: 13 Dec 2017

Author’s response to reviews:

We have revised the section regarding the authors contribution to this manuscript. I hereby declare that all authors made and undisputed and substantive intellectual contribution to this study and meet the criteria of the ICMJE guidelines.

In addition I would like to remind the editor that the submission process of this manuscript has now entered its second year. If, after this revision, the editor still feels the need for further - non-relevant - revisions, I will have to withdraw the manuscript in order to prevent it from becoming irrelevant due to the time passed.